Author's response to reviews

Title: eCROPS-CA: a systematic approach toward effective and sustainable cancer prevention in rural China

Authors:

Jing Chai (jingchai82@sina.com)
Xingrong Shen (1264854421@qq.com)
Rui Feng (853571123@qq.com)
Jing Chen (jing_hp@sina.com)
Han Liang (1010066583@qq.com)
Ting Zhao (1252509750@qq.com)
Rui Sha (1214984721@qq.com)
Yong Shi (295240260@qq.com)
Kaichun Li (13505612172@163.com)
Debin Wang (dbwang@vip.sina.com)

Version: 6
Date: 25 June 2014

Author's response to reviews:

Dear editor,

Thank you for your email regarding our submission to BMC Cancer. Our responses to the points and change requests raised in your email are as follows.

1. We have forwarded the requested documentation about ethical approval and funding approval to BMCSeriesEditorial@biomedcentral.com, including photo copies of the original Chinese versions and translated versions signed by contactable person at our university.

2. The study is funded mainly by the National Science Foundation of China and the study protocol had undergone peer-review by the funding body.

3. It is an ongoing study and we had just started recruiting patients.

4. We confirm that no results of this study have already been published or submitted to any journal, although it has benefited, as clearly indicated in the manuscript, greatly from two of our previously published studies.

5. We had asked help from an English teacher for correcting language problems with our manuscript.

With best regards,
Debin Wang
School of Health Service Management, Anhui Medical University, China